Author’s response to reviews

Title: Prevalence and associated factors for Pterygium in Han and Mongolian adults: acrosssectional study in inner Mongolian, China

Authors:
Yuhan Wang (drwangyh34@126.com)
Guangliang Shan (guangliang_shan@163.com)
Linyang Gan (ganly08@yeah.net)
Yonggang Qian (qyg0406@126.com)
Ting Chen (ct19870629@hotmail.com)
Hailing Wang (wanghailing1961@163.com)
Xiaodan Pan (1289905337@qq.com)
Wenrui Wang (wr821@163.com)
Li Pan (panli1716@163.com)
Xia Zhang (drzhangxia@bjmu.edu.cn)
Meng Wang (sophiawm27@126.com)
Jin Ma (majin1912@163.com)
Yong Zhong (zhongy_pumch@126.com)

Version: 2 Date: 14 Jan 2020

Author’s response to reviews:

Dear editor,

Thanks for your precise comments and we have modified our manuscript. The details of the modifications are as follows:

1. Ethics and consent: We have added the statements in.

2. Consent for publication: We have added ‘Not Applicable’ in this section.
3. Availability of data and material: We have added ‘Not Applicable’ in this section.

4. Role of Funders: We have supplied the declaration of funding in this section.

5. Authors' contributions: We have added the contributions of every author according the criteria for authorship in this section.

6. Remove supplementary file: We have removed the cover letter and STROBE guidelines checklist from the manuscripts.

7. Figure 1: The Figure 1 was drawn by the author YW, so we added the “Artificial drawing” in the legend.

8. Formatting: Thank you for your precise advice. We have added the Conclusions part in the discussion section.

9. Abbreviations: We have added the abbreviations according to the requirement.

10. Clean manuscript: We have deleted the tracked changes and highlights. Figures are uploaded as separate files.